Author's response to reviews

Title: Physical activity before and after breast cancer diagnosis and survival - The Norwegian Women and Cancer cohort study

Authors:

Kristin B Borch (kristin.benjaminsen.borch@uit.no)
Tonje Braaten (tonje.braaten@uit.no)
Eiliv Lund (eiliv.lund@uit.no)
Elisabete Weiderpass (elisabete.weiderpass.vainio@ki.se)

Version: 4 Date: 19 November 2015

Author's response to reviews:

In table 1 the total comes to 198 not 197. Is this a mistake or double counting?
Authors response: This is a mistake, and we have corrected other cancers to 26 deaths which is the correct number and the total now sums up to 197.